Author's response to reviews

Title: Systematic-Review Automation Technologies

Authors:

Guy Tsafnat (guyt@unsw.edu.au)
Paul Glasziou (pglaszio@bond.edu.au)
Miew Keen Choong (m.choong@unsw.edu.au)
Adam Dunn (a.dunn@unsw.edu.au)
Filippo Galgani (galganif@cse.unsw.edu.au)
Enrico Coiera (e.coiera@unsw.edu.au)

Version: 5  Date: 18 June 2014

Author's response to reviews: see over
Dear Systematic Reviews Editorial Team:

We have formatted the manuscript according to the guidelines for a Commentary (http://www.systematicreviewsjournal.com/authors/instructions/commentary) with the exception of the length restriction. Specifically, we have:

- changed the title to not include the word "survey" in it
- removed the structure formatting from the abstract
- renamed the Methods and Results & Discussion sections
- added a table of abbreviations.

We also corrected all references that showed up as Error! Reference source not found.

Best regards

Guy Tsafnat on behalf of all authors